Author's response to reviews

Title: Barbers' knowledge and practice about occupational biological hazards was low in Gondar town, North West Ethiopia

Authors:

Teresa Kisi Beyen (terek7@gmail.com)
Ketema Tafess Tulu (ketema_taf@yahoo.com)
Abdella Amano Abdo (a2ay_oro@gmail.com)
Abera Shibru Tulu (shibruabera65@gmail.com)

Version: 5 Date: 27 October 2012

Author's response to reviews: see over
To: Editorial Board  
BMC public Health  

27, October 2012

Dear sir/madam,

It is to be recalled that we have sent a paper entitled ‘Barbers’ knowledge and practice about biological hazards associated with their work was low in Gondar town, North West Ethiopia’ to be published on your journal. So, we would like to thanks a lot for considering our work to be published on you reputable journal. The comments have been addressed one by one as follow.

<table>
<thead>
<tr>
<th>Comment</th>
<th>Response to comment</th>
</tr>
</thead>
<tbody>
<tr>
<td>1. Competing interests</td>
<td>1. The authors declare that they have no competing interests.</td>
</tr>
<tr>
<td>2. figures</td>
<td>2. Comment was accepted and adjusted accordingly.</td>
</tr>
</tbody>
</table>